Author's response to reviews

Title: Anthracyclins, proteasome activity and mult-drug-resistance

Authors:

Mirela R Fekete (mirela22@hotmail.com)
William H McBride (wmcbride@mednet.ucla.edu)
Frank Pajonk (fpajonk@mednet.ucla.edu)

Version: 5 Date: 1 September 2005

Author's response to reviews:

The manuscript was revised according to the format guidelines of the Journal